Author's response to reviews

Title: A systematic review of effectiveness and safety of different regimens of levonorgestrel oral tablets for emergency contraception

Authors:

Mohammad Shohel (mohammadnsu@yahoo.com)
Mohammad M Rahman (dhakaiyamhfuz@yahoo.com)
Asif Zaman (asifzaman.nsu@gmail.com)
Mir MN Uddin (nasirmir5767@gmail.com)
MD M Al-Amin (bd_pharmacy@yahoo.com)
Hasan M Reza (rezami@hotmail.com)

Version: 6
Date: 27 March 2014

Author's response to reviews:

To, 25/03/2014
The Executive Editor,
BMC Women’s Health

Sub: Submission of revised manuscript

Dear Executive Editor,

I would like to take this opportunity to thank our reviewers for their valuable comments. They have raised some interesting issues which made us to think more critically. We have included Acknowledgement and sources of funding in our revised manuscript. Responses to Reviewer’s comments also enclosed with this cover letter. Based on reviewer’s comments, we have also revised our manuscript and the changes are giver below:

1. first line changed in abstract: Unintended pregnancy is a complex phenomenon which raise to take an emergency decision

2. Line added in Background section in page 3: But as data suggest, the significance of emergency contraceptive pills cannot be ignored any longer, since one million abortions and two million unwanted pregnancies could be prevented in the U.S. each year if its widespread use could be ensured

3. References were revised

We look forward to hearing from you at your earliest convenience.

Yours sincerely,

Mohammad Shohel
Lecturer
Department of Pharmaceutical Sciences
North South University